\font\bg=cmr17
\def\newsec#1{{\par\bigskip\bf #1:}}\footline={}
\parindent=0pt
\null\vskip -1.6cm
\centerline{\bg UNIVERSITY OF NOTRE DAME}
\smallskip
\centerline{DEPARTMENT OF MATHEMATICS}
\medskip
\centerline{\bg MATHEMATICS 105 {--} ELEMENTS OF CALCULUS I}
\smallskip
\centerline{{\bf Fall Semester 1991}}
\bigskip
\newsec{TEXT}
{\sl Calculus and its Applications\/}\ 
(Edition 5) by Goldstein, Lay and Schneider

\newdimen\B\B=6pt
\newsec{EXAMINATION SCHEDULE}
\medskip
\hskip 1in\vbox{{\settabs\+\hskip 1cm& Exam III&\hskip 1cm&Tuesday,  September 17&at 8:00 a.m.\cr
\vskip\B
\+& Exam I && Tuesday,  September 17&at 8:00 a.m.\cr
\vskip\B
\+& Exam II && Thursday, October  17&at 8:00 a.m.\cr
\vskip\B
\+& Exam III && Tuesday, November  26&at 8:15 a.m.\cr
\vskip\B
\+&Final && Monday, December 16&at 1:45 p.m.\cr}}
\medskip
The rooms for these exams will be announced in class 
as they are given to us by the Registrar.
These rooms will almost certainly NOT be your usual classroom 
and may very well be different for different sections.
They may also change from exam to exam at the discretion of the Registrar.
Exams I{--}III are each one hour long and each is worth 100 points.
The final exam lasts two hours and is worth 150 points.

\newsec{EXAMINATION PROCEDURES}
{{\parindent=24pt
Each student is expected to attend the exams at 
the scheduled times and in the assigned room. 
It is YOUR responsibility to do this.
Some of you will have legitimate excuses and conflicts which may
prevent this.
The sooner we know of these problems the more likely it is that
a satisfactory solution can be worked out.
Specifically, but not exclusively:
\itemitem{(i)}any conflicts between these examinations and other scheduled 
examinations you may have should be brought to the 
attention of your instructor as soon as they are known to you;
\itemitem{(ii)}the schedule of Finals is known NOW, so you should 
check to see that you do not have 3 examinations in a 24 hour period, etc.(if you do let your instructor know now);
\itemitem{(iii)}mail a copy of these dates to your parents so that 
there will not be family conflicts (your mother likes to hear from you anyway);
\itemitem{(iv)}if you miss a scheduled examination without 
an official excuse from the
Associate Vice President for Residence Life, you will receive a score of
0 for that examination;
\itemitem{(v)} if you do miss an exam, advise your instructor of the situation immediately.

\parindent=0pt
These exams are given under the provisions of the Honor Code {--} if you entered Notre Dame before
the introduction of the Honor Code and are unwilling to be bound by its provisions, contact your
instructor by September 6.

}}

\newsec{GRADES}
Your grade for this course will be determined primarily from 
your numerical scores on the four exams.
After each of the hour exams, we will post a letter grade scale of A,B,C,D or F, 
as an indication of how well you did on that exam.
The midterm grade for freshman will be A,B,C,D or F and will be 
based on your exam scores for the first two tests.
We will give pluses and minus for final grades.
The course grade itself is determined by your total test scores.
In case you are close to the border between two grades, your instructor
may use attendance, quiz scores, homework scores, etc.\ to decide your final
letter grade.

\newsec{ATTENDANCE}
Freshman are required by the University to attend all classes.
According to University regulations, a freshman may be given the grade of ``F'' if he or she
accumulates more than three unexcused absences after receiving a written warning from
the instructor.
All students are responsible for any information given out in class (e.g.\ exam room assignments 
and homework) and for any work done in class (e.g.\ quizzes).

\newsec{HOMEWORK}
Homework will be assigned regularly and it will be graded.
It should be done by all students.
The instructor may use homework records to decide borderline grade cases at the
end of the semester.
You should be sure you understand
your instructor's directions as to how homework will be assigned and collected.

\newsec{HELP}
There will be two help sessions per week, except for the week of an exam when special
arrangements will be made.
The normal schedule will be Tuesday and Thursday evenings beginning at 7:00 p.m.\ in
room 226 of the Math.\ building.

\end